

Health-related quality of life (HRQoL) after different axillary treatments in women with breast cancer: a 1-year longitudinal cohort study

N. J. M. C. Vrancken Peeters¹ · Z. L. R. Kaplan² · M. E. Clarijs¹ · M. A. M. Mureau³ · C. Verhoef¹ · T. van Dalen¹ · O. Husson^{1,4} · L. B. Koppert¹

Accepted: 3 October 2023 © The Author(s) 2023

Abstract

Purpose As life expectancy continues to rise, post-treatment health-related quality of life (HRQoL) of breast cancer patients becomes increasingly important. This study examined the one-year longitudinal relation between axillary treatments and physical, psychosocial, and sexual wellbeing and arm symptoms.

Methods Women diagnosed with breast cancer who received different axillary treatments being axilla preserving surgery (APS) with or without axillary radiotherapy or full axillary lymph node dissection (ALND) with or without axillary radiotherapy were included. HRQoL was assessed at baseline, 6- and 12-months postoperatively using the BREAST-Q and the European Organization for Research and Treatment of Cancer QoL Questionnaire Breast Cancer Module (EORTC QLQ-BR23). Mixed regression models were constructed to assess the impact of axillary treatment on HRQoL. HRQoL at baseline was compared to HRQoL at 6- and at 12-months postoperatively.

Results In total, 552 patients were included in the mixed regressions models. Except for ALND with axillary radiotherapy, no significant differences in physical and psychosocial wellbeing were found. Physical wellbeing decreased significantly between baseline and 6- and 12-months postoperatively (p < 0.001, p = 0.035) and psychosocial wellbeing decreased significantly between baseline and 12 months postoperatively (p = 0.028) for ALND with axillary radiotherapy compared to APS alone. Arm symptoms increased significantly between baseline and 6 months and between baseline and 12 months postoperatively (p = 0.028) for ALND with axillary radiotherapy compared to APS alone. Arm symptoms increased significantly between baseline and 6 months and between baseline and 12 months postoperatively for APS with radiotherapy (12.71, 13.73) and for ALND with radiotherapy (13.93, 16.14), with the lowest increase in arm symptoms for ALND without radiotherapy (6.85, 7.66), compared to APS alone (p < 0.05).

Conclusion Physical and psychosocial wellbeing decreased significantly for ALND with radiotherapy compared to APS alone. Shared decision making and expectation management pre-treatment could be strengthened by discussing arm symptoms per axillary treatment with the patient.

Plain English summary

Breast cancer is one of the most common cancers among women worldwide. Since the survival rates are improving, the health-related quality of life of breast cancer patients after treatment becomes more important. One of the most important factors in the prognosis of breast cancer patients is the presence of axillary metastasis, the spread of breast cancer to lymph nodes in the armpit. There are different axillary treatments which can all affect health-related quality of life negatively. Therefore, the aim of this study is to examine the effect of different axillary treatments on physical, psychosocial, and sexual wellbeing and arm symptoms of breast cancer patients. This study included women with breast cancer who received either

L. B. Koppert l.koppert@erasmusmc.nl

 Department of Surgical Oncology, Erasmus MC Cancer Institute, University Medical Center Rotterdam, 3015 GD Rotterdam, The Netherlands

- ² Department of Public Health, Erasmus MC, University Medical Center Rotterdam, Rotterdam, The Netherlands
- ³ Department of Plastic & Reconstructive Surgery, Erasmus MC Cancer Institute, University Medical Center Rotterdam, Rotterdam, The Netherlands
- ⁴ Department of Medical Oncology, Netherlands Cancer Institute-Antoni Van Leeuwenhoek, Amsterdam, The Netherlands

axillary preserving surgery with or without axillary radiotherapy, or a full axillary lymph node dissection with or without axillary radiotherapy. health-related quality of life was assessed with validated breast cancer specific questionnaires at baseline, 6- and 12-months postoperatively. Patients who underwent axilla preserving surgery without axillary radiotherapy experienced significantly less decline in both physical and psychosocial wellbeing as well as significantly fewer increase in arm symptoms compared to patients who received the more extensive axillary treatments. The results of this study can be used to improve shared decision making and expectation management for future breast cancer patients.b

Keywords Health-related quality of life (HRQoL) \cdot Patient-reported outcomes (PROs) \cdot Breast cancer \cdot Axilla preserving surgery (APS) \cdot Axillary lymph node dissection (ALND) \cdot Axillary radiotherapy

Background

Breast cancer is one of the most prevalent types of cancer among women, with an incidence of 16.000 new cases of invasive breast cancer per year in the Netherlands [1–3]. Due to the introduction of multiple novel therapies, the 10-year survival rate has gradually increased from 56 to 83% [4, 5]. Although the improvements in life expectancy are very promising, this also means that patients have to live with the possible negative consequences of breast cancer diagnosis and treatment which may affect their healthrelated quality of life (HRQoL), addressing the need for tailored treatment regimens [6, 7].

Axillary lymph node metastases are an important predictor of survival and recurrence in breast cancer patients and make treatment beyond the breast necessary [8-11]. In the early 90 s, the Sentinel Lymph Node Biopsy (SLNB) was introduced as a less invasive diagnostic for axillary lymph node staging in clinically node-negative (cN0) breast cancer patients [12]. If histopathological examination determined tumour-positive cells in the SLNB, several treatment options could follow, including an axillary lymph node dissection (ALND) and radiotherapy at different lymph node levels [13]. Treatment with ALND can be safely omitted in case of early stage breast cancer in a selection of patients who are treated with adjuvant chemotherapy or radiotherapy [14–17]. In comparison, clinically node-positive (cN+) patients were until recently, routinely treated with an ALND combined with or without axillary radiotherapy, also after neoadjuvant treatment, irrespective of their response to systemic treatment [13].

Axillary treatment with ALND and radiotherapy can cause significant upper extremity morbidities, such as disfunction of the arm, radiation fibrosis and lymphedema causing a decrease in HRQoL [18–20]. Recently, less invasive approaches for axillary treatment in cN + patients were introduced, known as Marking the Axilla with Radioactive Iodine seeds (MARI) procedure and Radioactive Iodine Seed Localization in the Axilla With the Sentinel Node (RISAS) procedure, a form of targeted axillary dissection [21–23]. The RISAS procedure is a combination of the SLNB and MARI procedure [22]. The marking procedure with iodine seeds is very elegant since, due to its long half lifetime, this marker stays radioactive during the months neo-adjuvant treatment is given. Additional treatment with ALND can be safely omitted when patients have a complete pathologic response according to the RISAS after neo-adjuvant treatment [21, 24]. Axillary preserving surgery has been associated with fewer complications compared to ALND and may, therefore, result in better HRQoL outcomes in breast cancer patients [16, 17, 25–27].

Previous studies have mainly focused on oncological outcomes of different axillary treatments in terms of overall survival or disease free survival [14–17, 28, 29]. Several studies have tried to investigate the effect of axillary treatment on HRQoL but lacked the inclusion of novel axillary preserving techniques or radiotherapy to the axilla. Additionally, the methodological quality shows wide variability with relative small sample sizes and lack the inclusion of baseline HRQoL measurements [26, 30–32]. Therefore, to enhance expectation management and shared decisionmaking, the objective of this study was to examine the longitudinal relation between different axillary treatments and HRQoL regarding the first year after surgery.

Methods

This is a longitudinal cohort study of women with breast cancer who received different axillary treatments for lymph node staging and/ or metastasis at the Erasmus MC Cancer Institute in the Netherlands between November 1st, 2015, and January 1st, 2022. Inclusion criteria were women with a diagnosis of breast cancer who received axillary treatment and who completed Patient-Reported Outcome Measures (PROMs) in the "patient data platform", Erasmus MC's online PROM collection tool [33]. Exclusion criteria were patients who underwent proton therapy, palliative treatment, patients with recurrent breast cancer or if data on treatment were missing.

Treatment

QoL was compared between four different axillary treatment groups, namely: [1] patients with axilla preserving surgery (APS) including SLNB and RISAS, [2] patients with an ALND (whether or not preceded by APS), [3] patients with APS followed by axillary radiotherapy and [4] patients with an ALND followed by axillary radiotherapy. In a SLNB, the median number of lymph nodes removed is 2. The RISAS procedure involves the removal of the marked lymph node metastasis plus the SLNB, which are in 71.3% the same lymph nodes. During an ALND usually a minimum of 10 lymph nodes are removed. The SLNB and RISAS procedure are combined in to one group since they only differ by a maximum of one lymph node in less than 30% of the cases.

Surgical treatment of the breast included either breastconserving surgery (BCS) or mastectomy with or without reconstruction. Chemotherapy, hormonal therapy, and/ or (loco)regional radiotherapy were given to patients according to the national guidelines for breast cancer.

Data-collection procedure

Data on sociodemographic factors, clinical treatment and tumour characteristics were retrospectively collected from medical records. These variables entailed, age at diagnosis, body mass index (BMI), gene mutation, presence of bilateral cancer, hormone receptor status, her2Neu receptor status and stage of disease at presentation (TNM staging system). Treatment characteristics entailed radiotherapy of the axilla and/or breast/ chest, parasternal radiotherapy, neo-adjuvant and/or adjuvant chemo(immuno)therapy, neo-adjuvant and/ or adjuvant hormonal therapy, type of breast surgery and type of axillary treatment.

HRQoL assessment

HRQoL data are routinely collected within the Erasmus MC Cancer Institute at baseline (first doctors visit; after diagnosis and before treatment initiation), at 6 months and at 12 months postoperatively according to the ICHOM Breast Cancer standard set [33, 34]. In this study, breast cancer specific questionnaires were used to assess HRQoL, namely the BREAST-Q and the EORTC QLQ-BR23 [35, 36]. In the statistical analysis only those domains in both questionnaires that are considered clinically the most relevant for comparing different axillary treatments were included to prevent the issues of multiple testing.

The BREAST-Q consists of two overarching themes, namely HRQoL and Patient Satisfaction. The HRQoL

domains include psychosocial, sexual, and physical wellbeing. The satisfaction domains include satisfaction with breasts, outcome, and care. Scores of each domain range from 0–100 with higher scores representing better HRQoL and satisfaction. A minimal important difference score of 4 points was considered clinically relevant [35, 37]. Domains on psychosocial, physical, and sexual wellbeing from the BREAST-Q were included in the statistical analysis.

The EORTC QLQ-BR23 is a breast cancer-specific questionnaire consisting of 23 items. It incorporates functional scales and breast cancer specific symptom scales. Each item is scored on an ordinal scale: 1 = not at all, 2 = a little, 3 = quite a bit, 4 = very much [36]. According to the EORTC scoring manual, the EORTC QLQ-BR23 scores were linearly transformed to a numeric score (range 0–100) [36, 38]. For functional scales, higher scores represent a better level of functioning. Within the symptom-oriented scales, higher scores represent more severe symptoms. A minimal important difference score of 5 points on the transformed 0 to 100 scale was considered clinically relevant [38, 39]. The domain on arm symptoms from the EORTC QLQ-BR23 was included in the statistical analysis.

Data analysis

Descriptive statistics, including frequencies and proportions, were used to describe patient, tumour, and treatment characteristics. Depending on the type of variable, ANOVA and Chi-square tests were used to compare the baseline characteristics of the four treatment groups. Means, standard deviations and percentages were calculated to present the results of HRQoL for each axillary treatment group. In addition, the percentages of missing values were also calculated. All scores were tested for normality with the Kolmogorov–Smirnov and Shapiro–Wilk tests. If normally distributed, parametric ANOVA tests were used to compare HRQoL between the pre-defined axillary treatment groups. If not normally distributed, non-parametric Kruskal–Wallis tests were used.

The longitudinal nature of the data, with repeated measurements within one patient (i.e., scores at baseline, 6- and 12- months postoperatively) induces correlation among these measurements and demands a multilevel approach. Therefore, to assess the effect of different axillary treatments on psychosocial wellbeing, physical wellbeing, sexual wellbeing, and arm symptoms, mixed effect linear regression was used. Patient was included as a random effect and the following fixed effects were adjusted for; age at diagnosis, BMI, type of breast surgery, radiotherapy of breast/chest, parasternal radiotherapy, neo-adjuvant hormonal therapy, neo-adjuvant chemo(immuno)therapy, postoperative hormonal therapy, and postoperative chemo(immuno)therapy. To avoid multicollinearity, tumour characteristics were not

Fig. 1 Flowchart



included as variables in the mixed models since they determine the treatment protocol. To analyse the impact of axillary treatment at different time points, an interaction term between time (baseline, 6 months, and 12 months) and axillary treatment was included in the model. APS without additional axillary radiotherapy was chosen as reference group since this is the least extensive axillary treatment. The effect of the different axillary treatments, between the different time periods, on the predefined outcomes was expressed as the delta. Delta 1 represents the change in HRQoL from the baseline measurement to 6 months postoperatively, and delta 2 represents the change in HRQoL from baseline to 12 months postoperatively. The delta gives more intuitive information compared to the crude measurements as it represents the relative effect of treatment on HRQoL. Positive deltas in BREAST-Q domains indicate an improvement in HRQoL, negative deltas a decline in HRQoL. Conversely, positive deltas in the arm symptoms domain of the EORTC QLQ-BR23 reflect an increase in arm symptoms and negative deltas a reduction in arm symptoms since this is a symptom orientated domain.

Non-responder analysis

Because not all patients in the patient data platform completed all PROMs, a non-responder analysis was conducted to examine possible significant differences between responders and non-responders. Student's unpaired t-tests and Chisquare tests were used to compare baseline characteristics. For all statistical analyses, a two-sided p-value of 5% was considered statistically significant. Statistical analyses were done using R statistical software (version 4.4.2) [40].

Sample size calculation

As this study is a secondary analysis on a healthcare registration process, a sample size calculation is not applicable in this context.

Results

Patient characteristics

A total of 734 breast cancer patients was included from the institutional database, of whom 731 patients remained after duplicates were removed. Of those patients, 152 were excluded because they did not meet the inclusion criteria. Another 27 non-responders (<5%) did not fill out the PROMs and were excluded, leaving 552 patients for the analyses (Fig. 1). Of the 552 included patients, 386 patients (69.9%) underwent APS, 59 patients (10.7%) an ALND, 39 patients (7.1%) APS in combination with axillary radiotherapy and 68 (12.3%) patients underwent an ALND in combination with axillary radiotherapy (Table 1).

Most baseline characteristics, except for BMI and neoadjuvant hormonal therapy, demonstrated significant differences between treatment groups. Patients who underwent ALND were more likely to have undergone a mastectomy, had a higher tumour stage, and received systemic therapy more frequently (Table 1).

Non-responder analysis

The number of non-responders in the study was 27 (<5%). The non-responder analysis showed similar distributions between responders and non-responders for almost all variables, except for BMI category. Non-responders had a higher BMI (BMI > 30) compared to responders (p=0.008). The full non-responder analysis can be found in Apendix 1.

HRQoL scores according to axillary treatment

There were no statistically significant differences in sexual wellbeing scores between treatment groups (Table 2). However, physical wellbeing scores differed significantly between treatment groups at 6 months postoperatively (p < 0.001, Table 2). Additionally, the psychosocial wellbeing domain also showed significant differences between treatment groups at 6 months (p = 0.010) and 12 months postoperatively (p = 0.012, Table 2). Furthermore, at 6- and 12-months postoperatively arm symptoms differed significantly between treatment groups (both p < 0.001, Table 2). HRQoL scores of other domains of the BREAST-Q and EORTC QLQ-BR23 can be found in Appendix 2.

Effect of axillary treatments on HRQoL over time

HRQoL measured at baseline did not significantly differ between the more extensive axillary treatment groups (APS with axillary radiotherapy, ALND and ALND with axillary radiotherapy) compared to the APS group (reference) after adjusting for all variables (Table 3). For patients who underwent APS, physical wellbeing, psychosocial wellbeing, and sexual wellbeing decreased, and arm symptoms increased significantly at 6- and 12 months postoperatively (p < 0.05, Table 3, Fig. 2).

For patients who received APS with axillary radiotherapy, delta 1 (i.e., the difference between baseline and 6 months postoperatively) and delta 2 (i.e., the difference between baseline and 12 months postoperatively) of the arm symptoms domain increased significantly with 12.71 and 13.74 points respectively compared to patients undergoing APS alone (p < 0.05, Table 3, Fig. 2d). Similarly, for patients who underwent ALND, delta 1 (6.85) and delta 2 (7.66) of the arm symptoms domain increased significantly compared to patients undergoing APS alone (p < 0.05, Table 3, Fig. 2d). This indicates that arm symptoms increased significantly over time for patients undergoing APS with axillary radiotherapy and for patients undergoing ALND compared to patients undergoing APS alone. Furthermore, the increase in arm symptoms is greater for patients receiving APS with axillary radiotherapy than for patients receiving ALND. There were no significant differences in other BREAST-Q domains (Table 3, Fig. 2).

For patients who received ALND combined with axillary radiotherapy, negative delta 1 of the physical wellbeing domain increased with 10.64 significantly compared to patients undergoing APS (p < 0.001, Table 3, Fig. 2a). The corresponding negative delta 2 increased significantly with 6.11 points (p = 0.035, Table 3, Fig. 2a). Moreover, negative delta 2 of the psychosocial wellbeing domain increased with 6.81 points significantly for patients receiving ALND with axillary radiotherapy compared to patients undergoing APS (p=0.028, Table 3, Fig. 2b). This indicates that the negative effect of axillary treatment on physical and psychosocial wellbeing increased significantly for patients undergoing ALND with axillary radiotherapy compared to patients undergoing APS alone. In addition, delta 1 (13.93) and delta 2 (16.14) of the arm symptoms domain increased significantly compared to patients undergoing APS (p < 0.001, Table 3, Fig. 2d). Sexual wellbeing did not significantly differ compared to APS (Table 3, Fig. 2c).

The effect plots demonstrate a decline in HRQoL from baseline to 6 months postoperatively. However, between 6 and 12 months postoperatively, HRQoL tends to stabilize again. The red colour represents the APS group, green the ALND group, blue the APS with axillary radiotherapy group and purple the ALND with axillary radiotherapy group (Fig. 2).

Discussion

This longitudinal cohort study of women with breast cancer aimed to investigate the association between different axillary treatments and HRQoL measured with the BREAST-Q and the EORTC QLQ-BR23 in the first year after surgery. The results demonstrated that all axillary treatments included in this study led to an increase in arm symptoms. Moreover, APS with axillary radiotherapy and ALND with and without axillary radiotherapy led to a significant greater increase in arm symptoms compared to APS alone. In addition, the most extensive axillary treatment consisting of ALND combined with axillary radiotherapy had also a clinically significant effect on physical and psychosocial wellbeing. This indicates

 Table 1
 Baseline characteristics according to treatment group

Characteristics	APS ^{<i>a</i>} $(n = 386)$	ALND $(n=59)$	$APS^a + Rtx^a (n=39)$	$ALND + Rtx^{a} (n = 68)$	<i>p</i> -value
Age					
Mean (SD)	55.5 (14.1)	50.9 (15.3)	50.0 (13.5)	48.4 (12.8)	< 0.001
BMI (%)					
<25	169 (43.8%)	30 (50.8%)	21 (53.8%)	35 (51.5%)	0.313
25–30	161 (41.7%)	18 (30.5%)	14 (35.9%)	20 (29.4%)	
> 30	56 (14.5%)	11 (18.6%)	4 (10.3%)	13 (19.1%)	
Gene mutation (%)					
No mutation	128 (33.2%)	33 (55.9%)	17 (43.6%)	41 (60.3%)	< 0.001
BRCA 1 mutation	35 (9.1%)	0 (0%)	3 (7.7%)	2 (2.9%)	
BRCA 2 mutation	20 (5.2%)	3 (5.1%)	0 (0%)	4 (5.9%)	
Other gene mutation	8 (2.1%)	2 (3.4%)	0 (0%)	1 (1.5%)	
Unknown	195 (50.5%)	21 (35.6%)	19 (48.7%)	20 (29.4%)	
Bilateral cancer (%)		()			
Yes	15 (3.9%)	8 (13.6%)	0 (0%)	4 (5.9%)	0.006
No	371 (96.1%)	51 (86.4%)	39 (100%)	64 (94.1%)	0.000
Surgery (%)	(///)				
Mastectomy	77 (19.9%)	31 (52.5%)	8 (20.5%)	45 (66.2%)	< 0.001
BCS ^a	211 (54.7%)	21 (35.6%)	27 (69.2%)	19 (27.9%)	100001
Mastectomy + reconstruction	98 (25 4%)	7 (11 9%)	4(10.3%)	4 (5 9%)	
Recentor (%)	90 (<u>2</u> 5.170)	/ (11.570)	(10.570)	(3.570)	
Triple negative	57 (14.8%)	4 (6.8%)	7 (17 9%)	17 (25.0%)	< 0.001
HFR2 positive ^{b}	28(7.3%)	9(15.3%)	4 (10.3%)	17(25.0%) 17(25.0%)	< 0.001
HR^{a} positive & HFR2 negative	258 (66.8%)	46 (78 0%)	(10.5%)	34(50.0%)	
Unknown	43(11.1%)	0 (0%)	0(0%)	0(0%)	
$T_{-stage}(\%)$	45 (11.176)	0 (070)	0(070)	0(0/0)	
nT0	24 (6 2%)	6 (10.2%)	6 (15 4%)	17 (25.0%)	< 0.001
pT0 pT1	24(0.2%)	0(10.2%)	0(15.7%)	18 (26.5%)	< 0.001
pT1 pT2	242(02.7%)	10(32.2%)	9(23.1%)	13(10.1%)	
p12 pT3	3(0.8%)	5(85%)	9(23.1%)	0(13.2%)	
p13 pT4	0(0%)	5(8.5%)	0(0%)	2(2.9%)	
pre	48(12.4%)	0(0/0)	0(0%)	2(2.9%) 9(13.2%)	
N stage (%)	40 (12.470)	2 (3.4%)	0(070)	9 (13.270)	
nNO	278 (07.0%)	12 (22.007)	6 (15 107)	25 (26 90)	< 0.001
pN0	378(97.970) 8(2197)	15 (22.0%)	0(13.4%)	25(50.8%) 17(25.0%)	< 0.001
pN2	0(2.1%)	43(70.3%)	0(0%)	17(23.0%) 10(27.0%)	
p1v2	0(0%)	1(1.7%)	0 (0%)	19 (27.9%) 7 (10.2%)	
p_{1NS}	0 (0%)	0 (0%)	0(0%)	7 (10.3%)	
Neodajuvani CTx* (%)	228 (85.00)	22 (54 201)	20(5120)	17 (25.00/)	< 0.001
NO Vec	528 (85.0%)	32 (34.2%)	20(31.5%)	17(23.0%)	< 0.001
$\frac{1}{1} \sum_{i=1}^{n} CT \left(\frac{1}{2} \left(\frac{1}{2} \right) \right)$	38 (13.0%)	27 (43.8%)	19 (48.7%)	51 (75.0%)	
Aajuvani CTx* (%)	224 (82.00)	26 ((1.06/)	22 (50.00)	22 (40 501)	-0.001
NO	324 (83.9%)	30 (01.0%)	23 (39.0%)	35 (48.5%) 25 (51.5%)	< 0.001
Yes $HT = \frac{1}{2} \left(\frac{1}{2} \right)$	62 (16.1%)	23 (39.0%)	16 (41.0%)	35 (51.5%)	
Neoadjuvant HIX [*] (%)	272 (06 68)	54 (01 50)	20 (100%)	(5.05.00)	0.1.40
INO Mar	3/3 (90.6%)	54 (91.5%)	39 (100%) 0 (0%)	00 (90.0%)	0.149
	13 (3.4%)	5 (8.5%)	0 (0%)	5 (4.4%)	
Adjuvant H1x [*] (%)	010 (54 - 54)	11 (10 (20)	0.(00.10)	25 (26.0%)	
No	219 (56.7%)	11 (18.6%)	9 (23.1%)	25 (36.8%)	< 0.001
Yes	167 (43.3%)	48 (81.4%)	30 (76.9%)	43 (63.2%)	
Breast/ chest Rtx ^u (%)					
No	174 (45.1%)	21 (35.6%)	0 (0%)	0 (0%)	< 0.001

able i (continued)								
Characteristics Yes	APS ^{<i>a</i>} ($n = 386$)	ALND $(n=59)$	$APS^a + Rtx^a (n=39)$	$ALND + Rtx^a (n = 68)$	<i>p</i> -value			
Yes	209 (54.1%)	35 (59.3%)	37 (94.9%)	50 (73.5%)				
Yes & parasternal	3 (0.8%)	3 (5.1%)	2 (5.1%)	18 (26.5%)				
Rtx ^a boost (%)								
No	302 (78.2%)	41 (69.5%)	23 (59.0%)	49 (72.1%)	0.032			
Yes	84 (21.8%)	18 (30.5%)	16 (41.0%)	19 (27.9%)				

Table 1 (continued)

Bold values indicate the *p*-value < 0.05

^{*a*}APS, axilla preserving surgery; Rtx, radiotherapy; BCS, breast conserving surgery; HR, hormone receptor; CTx, chemo(immune)therapy; HTx, hormonal therapy

^bHR+/HR- combined

that ALND with axillary radiotherapy affects not only arm morbidity, but also more general HRQoL domains. This may be due to the potential physical and emotional side effects associated with axillary treatments such as pain, emotional distress, anxiety, and uncertainty about the long-term effects of the treatment [41, 42]. No significant difference in sexual wellbeing were found between the different axillary treatment groups.

The results of this study are consistent with some previous studies that have demonstrated benefits of SLNB compared to ALND [26, 31, 32, 43-47]. For instance, Belmonte et al. reported less arm morbidity in patients receiving SLNB compared to ALND which supports the use of SLNB. However, they did not find a significant difference in general well-being measured with the disease-specific FACT-B+4and the generic short form 36 health survey between ALND and SLNB [31]. Similarly, Peintiger et al. showed that SLNB is associated with significant less arm and shoulder morbidity compared to ALND. In addition, the study demonstrated significant higher levels of pain on the symptom scale of the EORTC QLQ-C30 in the ALND group [32]. Moreover, the prospective cohort study of Dabakuyo et al. showed that patients undergoing SLNB score better on both the global health status from the EORTC QLQ-C30, and the arm symptoms from the EORTC QLQ-BR23 compared to patients undergoing ALND [26]. A more recent study of Appelgren et al. also demonstrated that arm morbidity is significantly worse in patients receiving ALND compared to SLNB and highlights the importance of de-escalation of axillary surgery [47]. In contrast, Kootstra et al. did not demonstrate any association between SLNB and a better HRQoL measured with the EORTC QLQ-C30 compared to ALND [30]. Several randomized controlled trials have been conducted and all reported better HRQoL with fewer postoperative complications in patients undergoing SLNB compared to patients undergoing ALND [43-46]. The discrepancies between previous HRQoL studies can be explained by differences in sample size and by the fact that different validated questionnaires were used to measure HRQoL.

The AMAROS trial demonstrated that axillary radiotherapy may be a feasible alternative to ALND in cN+breast cancer patients, with comparable axillary control and less morbidity. However, the AMAROS did not demonstrate significant differences in HRQoL measured with the pain domain of the EORTC QLQ-C30 and the arm symptoms and body image domain of the EORTC QLQ-BR23 [17]. An interesting finding of the current study is that arm symptoms increased with 12.71 (delta 1) and 13.74 (delta 2) for patients undergoing APS with axillary radiotherapy compared to APS alone, while arm symptoms increased with 6.85 (delta 1) and 7.66 (delta 2) for patients undergoing ALND compared to APS alone. This suggests that patients who undergo APS with axillary radiotherapy may experience a greater increase in arm symptoms than patients undergoing an ALND. However, it is important to note that the comparison is not directly between APS with axillary radiotherapy and ALND. Therefore, more research is necessary to determine whether the difference in arm symptoms is statistically significant between these two axillary treatments.

Strengths and limitations

One of the main strengths of this study is its relatively large sample size and the availability of a wide range of patient, treatment, and tumour characteristics that were collected from medical records and used in the regression analysis. Furthermore, there are no previous studies that have compared HRQoL as measured by the BREAST-Q or EORTC QLQ-BR23 between the four different axillary treatment groups (including axillary radiotherapy). This makes this research valuable for future comparisons. Another major strength is the longitudinal design of the study. Baseline and postoperative HRQoL outcomes were included, which allowed for the observation of the effect of different axillary treatments on HRQoL over time while correcting for clinically relevant variables. Additionally, by including baseline HRQoL measurements, it was possible to account for any potential missing patient history.

Table 2 BREAST-Q and EORTC QLQ-BR23 scores on all follow-up times according to treatment group

HRQoL	APS	ALND	APS+Rtx	ALND+Rtx	<i>p</i> -value
	(n = 386)	(<i>n</i> =59)	(n=39)	(N=68)	
BREAST-Q: Physical wellbeing at baseline					
Mean (SD)	80.9 (15.2)	78.1 (14.8)	82.1 (15.4)	78.6 (17.8)	0.520
Missing (%)	89 (23.1%)	15 (25.4%)	7 (17.9%)	14 (20.6%)	
BREAST-Q: Physical wellbeing at 6 months					
Mean (SD)	69.4 (18.1)	66.8 (24.4)	67.1 (16.3)	57.8 (18.5)	< 0.001
Missing (%)	99 (25.6%)	12 (20.3%)	11 (28.2%)	16 (23.5%)	
BREAST-Q: Physical wellbeing at 12 months					
Mean (SD)	71.9 (20.4)	66.0 (26.7)	70.7 (21.2)	65.3 (20.6)	0.145
Missing (%)	87 (22.5%)	9 (15.3%)	8 (20.5%)	16 (23.5%)	
BREAST-Q: Psychosocial wellbeing at baseline					
Mean (SD)	74.2 (17.5)	70.1 (19.0)	75.2 (17.6)	72.3 (20.1)	0.510
Missing (%)	89 (23.1%)	15 (25.4%)	7 (17.9%)	14 (20.6%)	
BREAST-Q: Psychosocial wellbeing at 6 months					
Mean (SD)	69.8 (18.0)	61.9 (16.5)	63.8 (15.2)	64.0 (20.7)	0.010
Missing (%)	170 (44.0%)	19 (32.2%)	18 (46.2%)	22 (32.4%)	
BREAST-Q: Psychosocial wellbeing at 12 months					
Mean (SD)	68.9 (19.5)	61.3 (18.1)	65.1 (18.7)	62.0 (21.3)	0.012
Missing (%)	164 (42.5%)	14 (23.7%)	15 (38.5%)	21 (30.9%)	
BREAST-Q: Sexual wellbeing at baseline					
Mean (SD)	63.7 (20.9)	64.3 (23.0)	68.5 (17.4)	64.3 (21.3)	0.729
Missing (%)	151 (39.1%)	29 (49.2%)	15 (38.5%)	22 (32.4%)	
BREAST-Q: Sexual wellbeing at 6 months					
Mean (SD)	48.3 (17.8)	45.4 (17.3)	49.6 (12.3)	49.4 (15.7)	0.729
Missing (%)	176 (45.6%)	28 (47.5%)	22 (56.4%)	32 (47.1%)	
BREAST-Q: Sexual wellbeing at 12 months					
Mean (SD)	50.3 (19.6)	46.3 (18.7)	48.6 (16.4)	49.1 (21.0)	0.739
Missing (%)	171 (44.3%)	21 (35.6%)	21 (53.8%)	30 (44.1%)	
EORTC QLQ_BR2 ^a 3: Arm symptoms at baseline					
Mean (SD)	8.65 (15.8)	7.41 (13.6)	5.05 (7.90)	8.05 (16.0)	0.844
Missing (%)	84 (21.8%)	14 (23.7%)	6 (15.4%)	10 (14.7%)	
EORTC QLQ_BR23 ^a : Arm symptoms at 6 months					
Mean (SD)	13.2 (17.4)	19.0 (22.3)	22.6 (18.5)	26.6 (20.1)	< 0.001
Missing (%)	88 (22.8%)	11 (18.6%)	8 (20.5%)	12 (17.6%)	
EORTC QLQ_BR23 ^a : Arm symptoms at 12 months					
Mean (SD)	12.8 (17.2)	19.4 (21.6)	22.2 (18.9)	27.3 (20.7)	< 0.001
Missing (%)	73 (18.9%)	8 (13.6%)	7 (17.9%)	15 (22.1%)	

Bold values indicate the *p*-value < 0.05

Symptom-oriented scales; higher scores represent greater symptom severity

^aFunctional scales; higher scores represent a better level of functioning

An important limitation is the fact that patients were not randomly assigned to a treatment group since there are specific treatment protocols for each breast cancer stage [11]. This may lead to confounding by indication. Moreover, most of the baseline characteristics differed significantly between treatment groups. By including various patient and treatment characteristics in the mixed models, it was tried to correct for as many confounders as possible. Especially the correction for neo-adjuvant chemotherapy was important, as patients undergoing a RISAS procedure always undergo neoadjuvant chemotherapy in contrast to patients undergoing an SLND and because chemotherapy affects HRQoL [48]. There may be residual confounding since it is not possible to control for unknown confounders. Furthermore, there was no data on for example socioeconomic status, educational level, and coping strategies, which can also have an

Table 3 Mixed model regression analysis

	Physical wellbeing		Psychosocial wellbeing		Sexual wellbeing		Arm symptoms	
	Estimates	<i>p</i> -value	Estimates	<i>p</i> -value	Estimates	<i>p</i> -value	Estimates	<i>p</i> -value
APS: baseline (intercept)	66.29	< 0.001	61.60	< 0.001	67.30	< 0.001	18.78	< 0.001
APS: 6 months postoperatively	- 11.50	< 0.001	- 3.32	0.010	- 15.68	< 0.001	4.62	< 0.001
APS: 12 months postoperatively	-8.88	< 0.001	- 3.69	0.004	- 14.09	< 0.001	3.64	0.002
APS+Rtx	3.58	0.315	0.81	0.818	4.30	0.316	- 3.98	0.230
ALND	0.42	0.892	- 1.99	0.514	- 1.23	0.744	- 1.48	0.610
ALND+Rtx	4.37	0.193	2.92	0.380	2.44	0.526	- 1.98	0.516
$APS + Rtx: delta 1^{a}$	- 2.24	0.541	- 6.14	0.142	- 4.62	0.356	12.71	0.001
APS + Rtx: delta 2b	- 1.66	0.644	-4.20	0.300	- 6.31	0.199	13.74	< 0.001
ALND: delta 1 ^a	- 1.18	0.703	- 3.40	0.313	- 1.42	0.734	6.85	0.030
ALND: delta 2 ^b	- 3.56	0.239	- 3.53	0.276	- 1.15	0.773	7.66	0.014
ALND+Rtx: delta 1 ^a	- 10.64	< 0.001	- 5.10	0.104	0.01	0.998	13.93	< 0.001
ALND+Rtx: delta 2 ^b	- 6.11	0.035	- 6.81	0.028	- 1.05	0.774	16.14	< 0.001

Bold values indicate the *p*-value < 0.05

^aChange in HRQoL from baseline to 6 months postoperatively

^bChange in HRQoL from baseline to 12 months postoperatively



Treatment 🤗 APS = ALND = APS + Rtx = ALND + Rtx

Fig.2 Effect plots of HRQoL over time according to axillary treatment group with the corresponding confidence intervals. The following characteristics were used in the effect plots: mean age, BMI < 25, BCS, adjuvant hormonal therapy and radiotherapy of the breast. Red represents APS, green ALND, blue APS+Rtx and purple ALND+Rtx. **a** Physical wellbeing; **b** Psychosocial wellbeing; **c** Sexual wellbeing; **d** Arm symptoms impact on HRQoL [49, 50]. Additionally, the non-responder analysis revealed similar distribution patterns of baseline characteristics among the treatment groups, except for BMI. Non-responders tended to have a higher BMI compared to responders. Although BMI is linked to HRQoL, its influence on the overall outcomes is not expected to be significant since the non-responder subgroup accounts for less than 5% of the total sample size [51]. Lastly, there was a relatively high level of missingness per timepoint. This may be caused by patients switching hospitals or lacking motivation to complete PROMs. In addition, the sexual wellbeing domain of the BREAST-Q was not mandatory, resulting in even more missingness for this specific domain. The substantial missingness of data, especially in the BREAST-Q domains, may impact the validity of the results and may introduce bias. Therefore, it is recommended to validate the current results in other breast cancer cohorts.

Future perspectives

In the future, it would be beneficial to conduct a validation study utilizing a multicentre database such as the Dutch MINIMAX study to compare HRQoL among patients receiving APS with axillary radiotherapy and patients receiving ALND alone [52]. Those results will be of great value to both patients and healthcare providers especially when there is a choice between treatments with comparable oncological outcome. Moreover, it would be interesting to examine HRQoL at intervals of 2- or 5-years postoperatively to determine if there are any changes in HRQoL between treatment groups. Radiation fibrosis should then be included in the analysis as it typically occurs 12 months after treatment and significantly impacts HRQoL [53-55]. Additionally, the use of more fixed HRQoL outcomes defined in core outcome sets is recommended to make more meaningful comparisons in future studies.

The findings of this study will be informative for healthcare providers and breast cancer patients in the future as they will enhance expectation management and shared decision making. In addition, this study offers patients perspective and demonstrates that the decline in their HRQoL may stabilize again after six months. This has the potential to motivate patients to complete PROMs, which may result in more HRQoL data for future research. Furthermore, this study provides a foundation for further research on the impact of axillary treatment on HRQoL, as axillary treatment is an emerging area of research [16, 56–60]. Once the oncological safety of less invasive and less extensive axillary treatments has been evaluated and protocols may be adjusted, this study can be used to develop predictive models and decision-making tools for patients.

Conclusion

Physical and psychosocial wellbeing decreased significantly in patients receiving ALND with axillary radiotherapy compared to patients receiving APS alone. Patients undergoing APS without axillary radiotherapy had significantly fewer arm symptoms compared to patients receiving APS with axillary radiotherapy and patients receiving ALND with or without axillary radiotherapy. Shared decision making and expectation management could be strengthened by discussing arm symptoms per axillary treatment with the patient.

Supplementary Information The online version contains supplementary material available at https://doi.org/10.1007/s11136-023-03538-3.

Acknowledgements The authors wish to thank the Netherlands Institute for Health Sciences (NIHES) for supporting this research project.

Author contributions All authors contributed to the study conception and design. Material preparation, data collection and analysis were performed by NJMCVP, ZLRK and MEC. The first draft of the manuscript was written by NJMCVP and all authors commented on previous versions of the manuscript. All authors read and approved the final manuscript.

Funding The authors declare that no funds, grants, or other support were received during the preparation of this manuscript.

Data availability The datasets generated during and/or analysed during the current study are not publicly available due to privacy or ethical restrictions but are available from the corresponding author on reasonable request.

Declarations

Conflict of interest The authors have no relevant financial or non-financial interests to disclose.

Ethical approval This is an observational study. The Erasmus MC Research Ethics Committee has confirmed that no ethical approval is required.

Consent to participate The Institutional Review Board was consulted and concluded that informed consent was not needed since the valuebased healthcare-strategy is considered standard of care in Erasmus MC.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- Ghoncheh, M., Pournamdar, Z., & Salehiniya, H. (2016). Incidence and mortality and epidemiology of breast cancer in the world. *Asian Pacific journal of cancer prevention : APJCP*, *17*(S3), 43–46. https://doi.org/10.7314/apjcp.2016.17.s3.43
- World Health Organization. (2020). WHO. Early diagnosis and screening. 2015. Retrieved from https://www.who.int/cancer/ prevention/diagnosis-screening/breast-cancer/en/
- Volksgezondheidenzorg. (n.d.). Borstkanker sterfte en overleving. 13–09–2021. Retrieved February 2, 2020, from https://www. volksgezondheidenzorg.info/onderwerp/borstkanker/cijfers-conte xt/sterfte-en-overleving#node-sterfte-borstkanker
- (IKNL), I. K. N. (2020). Borstkanker in Nederland 1989–2017: hogere incidentie; betere overleving. Retrieved from https://iknl. nl/nieuws/2020/borstkanker-in-nederland-1989-2017-hogereincident
- Burguin, A., Diorio, C., & Durocher, F. (2021). Breast cancer treatments: Updates and new challenges. *Journal of personalized medicine*. https://doi.org/10.3390/jpm11080808
- JavanBiparva, A., Raoofi, S., Rafiei, S., PashazadehKan, F., Kazerooni, M., Bagheribayati, F., & Ghashghaee, A. (2022). Global quality of life in breast cancer: Systematic review and meta-analysis. *BMJ supportive & palliative care, bmjspcare*. https://doi.org/10.1136/bmjspcare-2022-003642
- Epplein, M., Zheng, Y., Zheng, W., Chen, Z., Gu, K., Penson, D., & Shu, X.-O. (2011). Quality of life after breast cancer diagnosis and survival. *Journal of clinical oncology : official journal of the American Society of Clinical Oncology, 29*(4), 406–412. https:// doi.org/10.1200/JCO.2010.30.6951
- 8. National Institute for Public Health and the Environment. (2018). Breast cancer in the Netherlands.
- Chang, J. M., Leung, J. W. T., Moy, L., Ha, S. M., & Moon, W. K. (2020). Axillary nodal evaluation in breast cancer: State of the art. *Radiology*, 295(3), 500–515. https://doi.org/10.1148/radiol. 2020192534
- 10 Weigelt, B., Peterse, J. L., & van Veer, L. J. (2005). Breast cancer metastasis: markers and models. *Nature reviews. Cancer*, 5(8), 591–602. https://doi.org/10.1038/nrc1670
- Federatie Medisch Specialisten. (2022). borstkanker richtlijnen. Retrieved from https://richtlijnendatabase.nl/richtlijn/borstkanker/ algemeen.html
- 12 Krag, D., Weaver, D., Ashikaga, T., Moffat, F., Klimberg, V. S., Shriver, C., & Beitsch, P. (1998). The sentinel node in breast cancer–a multicenter validation study. *The New England journal* of medicine, 339(14), 941–946. https://doi.org/10.1056/NEJM1 99810013391401
- 13. Nationaal Borstkanker Overleg Nederland. (2020). *Borstkanker richtlijn*.
- Li, C. Z., Zhang, P., Li, R. W., Wu, C. T., Zhang, X. P., & Zhu, H. C. (2015). Axillary lymph node dissection versus sentinel lymph node biopsy alone for early breast cancer with sentinel node metastasis: A meta-analysis. *European journal of surgical oncol*ogy: The journal of the European Society of Surgical Oncology and the British Association of Surgical Oncology, 41(8), 958–966. https://doi.org/10.1016/j.ejso.2015.05.007
- 15. Giuliano, A. E., McCall, L., Beitsch, P., Whitworth, P. W., Blumencranz, P., Leitch, A. M., & Ballman, K. (2010). Locoregional recurrence after sentinel lymph node dissection with or without axillary dissection in patients with sentinel lymph node metastases: the American college of surgeons oncology group z0011 randomized trial. *Annals of surgery*, 252(3), 426–432. https://doi. org/10.1097/SLA.0b013e3181f08f32
- 16 Giuliano, A. E., Ballman, K. V., McCall, L., Beitsch, P. D., Brennan, M. B., Kelemen, P. R., & Morrow, M. (2017). Effect of

axillary dissection vs no axillary dissection on 10-year overall survival among women with invasive breast cancer and sentinel node metastasis: the ACOSOG Z0011 (Alliance) randomized clinical trial. *JAMA*, *318*(10), 918–926. https://doi.org/10.1001/jama.2017.11470

- 17 Donker, M., van Tienhoven, G., Straver, M. E., Meijnen, P., van de Velde, C. J. H., Mansel, R. E., & Rutgers, E. J. T. (2014). Radiotherapy or surgery of the axilla after a positive sentinel node in breast cancer (EORTC 10981–22023 AMAROS): A randomised, multicentre, open-label, phase 3 non-inferiority trial. *The Lancet. Oncology*, *15*(12), 1303–1310. https://doi.org/10.1016/S1470-2045(14)70460-7
- 18 Petrek, J. A., Pressman, P. I., & Smith, R. A. (2000). Lymphedema: current issues in research and management. *CA a cancer journal for clinicians*, 50(5), 292–307. https://doi.org/10.3322/canjclin. 50.5
- 19 Senkus-Konefka, E., & Jassem, J. (2006). Complications of breast-cancer radiotherapy. *Clinical oncology (Royal College of Radiologists (Great Britain))*, 18(3), 229–235. https://doi.org/10. 1016/j.clon.2005.11.004
- Kwan, W., Jackson, J., Weir, L. M., Dingee, C., McGregor, G., & Olivotto, I. A. (2002). Chronic arm morbidity after curative breast cancer treatment: Prevalence and impact on quality of life. *Journal* of clinical oncology : Official journal of the American Society of Clinical Oncology, 20(20), 4242–4248. https://doi.org/10.1200/ JCO.2002.09.018
- 21 van Nijnatten, T. J. A., Simons, J. M., Smidt, M. L., van der Pol, C. C., van Diest, P. J., Jager, A., & Luiten, E. J. T. (2017). A novel less-invasive approach for axillary staging after neoadjuvant chemotherapy in patients with axillary node-positive breast cancer by combining radioactive iodine seed localization in the axilla with the sentinel node procedure (RISAS): A Dutch pros. *Clinical breast cancer*, *17*(5), 399–402. https://doi.org/10.1016/j. clbc.2017.04.006
- 22. Straver, M. E., Loo, C. E., Alderliesten, T., Rutgers, E. J. T., & Vrancken Peeters, M. T. F. D. (2010). Marking the axilla with radioactive iodine seeds (MARI procedure) may reduce the need for axillary dissection after neoadjuvant chemotherapy for breast cancer. *The British journal of surgery*, 97(8), 1226–1231. https:// doi.org/10.1002/bjs.7073
- Donker, M., Straver, M. E., Wesseling, J., Loo, C. E., Schot, M., Drukker, C. A., & VranckenPeeters, M.-J.T.F.D. (2015). Marking axillary lymph nodes with radioactive iodine seeds for axillary staging after neoadjuvant systemic treatment in breast cancer patients: the MARI procedure. *Annals of surgery*, 261(2), 378– 382. https://doi.org/10.1097/SLA.00000000000558
- Simons, J., Nijnatten, T. JAv., Koppert, L. B., van der Pol, C. C., Diest, PJv., Jager, A., Klaveren, Dv., Kam, B. L. R., Lobbes, M. B. I., de Boer, M., Verhoef, C., Sars, P. R. A., Heijmans, H. J., van Haaren, E. R. M., Vles, W. J., & Car, E. J. L. (2021). Abstract GS1–10: Radioactive Iodine seed placement in the axilla with sentinel lymph node biopsy after neoadjuvant chemotherapy in breast cancer: Results of the prospective multicenter RISAS trial. *Cancer Research*, *81*(Suppl 4), 10–10.
- Langer, I., Guller, U., Berclaz, G., Koechli, O. R., Schaer, G., Fehr, M. K., & Zuber, M. (2007). Morbidity of sentinel lymph node biopsy (SLN) alone versus SLN and completion axillary lymph node dissection after breast cancer surgery: a prospective Swiss multicenter study on 659 patients. *Annals of surgery*, 245(3), 452–461. https://doi.org/10.1097/01.sla.0000245472. 47748.ec
- 26. Dabakuyo, T. S., Fraisse, J., Causeret, S., Gouy, S., Padeano, M.-M., Loustalot, C., & Bonnetain, F. (2009). A multicenter cohort study to compare quality of life in breast cancer patients according to sentinel lymph node biopsy or axillary lymph node dissection. Annals of oncology : official journal of the European

Society for Medical Oncology, 20(8), 1352–1361. https://doi.org/ 10.1093/annonc/mdp016

- Ashikaga, T., Krag, D. N., Land, S. R., Julian, T. B., Anderson, S. J., Brown, A. M., National Surgical Adjuvant Breast, B. P. (2010). Morbidity results from the NSABP B-32 trial comparing sentinel lymph node dissection versus axillary dissection. *Journal of surgical oncology*, *102*(2), 111–118. https://doi.org/10.1002/jso.21535
- Reimer, T., Stachs, A., Nekljudova, V., Loibl, S., Hartmann, S., Wolter, K., & Gerber, B. (2017). Restricted axillary staging in clinically and sonographically node-negative early invasive breast cancer (c/iT1–2) in the context of breast conserving therapy: First results following commencement of the intergroup-sentinelmamma (INSEMA) trial. *Geburtshilfe und Frauenheilkunde*, 77(2), 149–157. https://doi.org/10.1055/s-0042-122853
- Kuemmel, S., Heil, J., Bruzas, S., Breit, E., Schindowski, D., Harrach, H., & Reinisch, M. (2023). Safety of targeted axillary dissection after neoadjuvant therapy in patients with node-positive breast cancer. *JAMA Surgery*. https://doi.org/10.1001/jamasurg. 2023.1772
- Kootstra, J., Hoekstra-Weebers, J. E. H. M., Rietman, H., de Vries, J., Baas, P., Geertzen, J. H. B., & Hoekstra, H. J. (2008). Quality of life after sentinel lymph node biopsy or axillary lymph node dissection in stage I/II breast cancer patients: A prospective longitudinal study. *Annals of surgical oncology*, 15(9), 2533–2541. https://doi.org/10.1245/s10434-008-9996-9
- Belmonte, R., Garin, O., Segura, M., Pont, A., Escalada, F., & Ferrer, M. (2012). Quality-of-life impact of sentinel lymph node biopsy versus axillary lymph node dissection in breast cancer patients. *Value in health : the journal of the International Society for Pharmacoeconomics and Outcomes Research*, *15*(6), 907–915. https://doi.org/10.1016/j.jval.2012.06.003
- 32. Peintinger, F., Reitsamer, R., Stranzl, H., & Ralph, G. (2003). Comparison of quality of life and arm complaints after axillary lymph node dissection vs sentinel lymph node biopsy in breast cancer patients. *British journal of cancer*, 89(4), 648–652. https:// doi.org/10.1038/sj.bjc.6601150
- 33. van Egdom, L. S. E., Lagendijk, M., van der Kemp, M. H., van Dam, J. H., Mureau, M. A. M., Hazelzet, J. A., & Koppert, L. B. (2019). Implementation of value based breast cancer care. *European journal of surgical oncology : The journal of the European Society of Surgical Oncology and the British Association of Surgical Oncology*, 45(7), 1163–1170. https://doi.org/10.1016/j.ejso. 2019.01.007
- 34. Ong, W. L., Schouwenburg, M. G., van Bommel, A. C. M., Stowell, C., Allison, K. H., Benn, K. E., & Saunders, C. (2017). A standard set of value-based patient-centered outcomes for breast cancer: The international consortium for health outcomes measurement (ICHOM) initiative. *JAMA oncology*, 3(5), 677–685. https://doi.org/10.1001/jamaoncol.2016.4851
- Pusic, A. L., Klassen, A. F., Scott, A. M., Klok, J. A., Cordeiro, P. G., & Cano, S. J. (2009). Development of a new patient-reported outcome measure for breast surgery: The BREAST-Q. *Plastic and reconstructive surgery*, *124*(2), 345–353. https://doi.org/10.1097/ PRS.0b013e3181aee807
- 36. Sprangers, M. A., Groenvold, M., Arraras, J. I., Franklin, J., te Velde, A., Muller, M., & Aaronson, N. K. (1996). The European organization for research and treatment of cancer breast cancerspecific quality-of-life questionnaire module: First results from a three-country field study. *Journal of clinical oncology : official journal of the American Society of Clinical Oncology, 14*(10), 2756–2768. https://doi.org/10.1200/JCO.1996.14.10.2756
- Voineskos, S. H., Klassen, A. F., Cano, S. J., Pusic, A. L., & Gibbons, C. J. (2020). Giving meaning to differences in BREAST-Q scores: Minimal important difference for breast reconstruction patients. *Plastic and reconstructive surgery*, *145*(1), 11e–20e. https://doi.org/10.1097/PRS.0000000006317

- Fayers P. M., Aaronson N. K., Bjordal K., Groenvold M., Curran D. B. A. (2001). The EORTC QLQ – C30 Scoring Manual 3rd edition. European Organization for Research and Treatment of Cancer.
- Osoba, D., Rodrigues, G., Myles, J., Zee, B., & Pater, J. (1998). Interpreting the significance of changes in health-related qualityof-life scores. *Journal of clinical oncology : Official journal of the American Society of Clinical Oncology, 16*(1), 139–144. https:// doi.org/10.1200/JCO.1998.16.1.139
- R Core Team (2022). R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. URL https://www.R-project.org/. (n.d.).
- Iddrisu, M., Aziato, L., & Dedey, F. (2020). Psychological and physical effects of breast cancer diagnosis and treatment on young Ghanaian women: A qualitative study. *BMC Psychiatry*, 20(1), 353. https://doi.org/10.1186/s12888-020-02760-4
- Lovelace, D. L., McDaniel, L. R., & Golden, D. (2019). Longterm effects of breast cancer surgery, treatment, and survivor care. *Journal of midwifery & women's health*, 64(6), 713–724. https:// doi.org/10.1111/jmwh.13012
- 43 Veronesi, U., Paganelli, G., Viale, G., Luini, A., Zurrida, S., Galimberti, V., & Fernández, J. R. (2006). Sentinel-lymphnode biopsy as a staging procedure in breast cancer: update of a randomised controlled study. *The Lancet. Oncology*, 7(12), 983–990. https://doi.org/10.1016/S1470-2045(06)70947-0
- 44 Fleissig, A., Fallowfield, L. J., Langridge, C. I., Johnson, L., Newcombe, R. G., Dixon, J. M., & Mansel, R. E. (2006). Postoperative arm morbidity and quality of life. Results of the ALMANAC randomised trial comparing sentinel node biopsy with standard axillary treatment in the management of patients with early breast cancer. *Breast cancer research and treatment*, 95(3), 279–293. https://doi.org/10.1007/s10549-005-9025-7
- 45. Purushotham, A. D., Upponi, S., Klevesath, M. B., Bobrow, L., Millar, K., Myles, J. P., & Duffy, S. W. (2005). Morbidity after sentinel lymph node biopsy in primary breast cancer: Results from a randomized controlled trial. *Journal of clinical oncology : Official Journal of the American Society of Clinical Oncology, 23*(19), 4312–4321. https://doi.org/10.1200/JCO. 2005.03.228
- Mansel, R. E., Fallowfield, L., Kissin, M., Goyal, A., Newcombe, R. G., Dixon, J. M., & Ell, P. J. (2006). Randomized multicenter trial of sentinel node biopsy versus standard axillary treatment in operable breast cancer: the ALMANAC Trial. *Journal of the National Cancer Institute*, 98(9), 599–609. https://doi.org/10. 1093/jnci/djj158
- Appelgren, M., Sackey, H., Wengström, Y., Johansson, K., Ahlgren, J., Andersson, Y., SENOMAC Trialists' Group. (2022). Patient-reported outcomes one year after positive sentinel lymph node biopsy with or without axillary lymph node dissection in the randomized SENOMAC trial. *Breast (Edinburgh, Scotland), 63*, 16–23. https://doi.org/10.1016/j.breast.2022.02.013
- Hwang, S. Y., Chang, S. J., & Park, B.-W. (2013). Does chemotherapy really affect the quality of life of women with breast cancer? *Journal of breast cancer*, *16*(2), 229–235. https://doi.org/10. 4048/jbc.2013.16.2.229
- 49 Sharma, N., & Purkayastha, A. (2017). Factors affecting quality of life in breast cancer patients: A descriptive and cross-sectional study with review of literature. *Journal of mid-life health*, 8(2), 75–83. https://doi.org/10.4103/jmh.JMH_15_17
- Ashing-Giwa, K. T., & Lim, J. (2009). Examining the impact of socioeconomic status and socioecologic stress on physical and mental health quality of life among breast cancer survivors. *Oncology nursing forum*, 36(1), 79–88. https://doi.org/10.1188/ 09.ONF.79-88
- 51. Pimenta, F. B. C., Bertrand, E., Mograbi, D. C., Shinohara, H., & Landeira-Fernandez, J. (2015). The relationship between obesity

and quality of life in Brazilian adults. *Frontiers in psychology*, 6, 966. https://doi.org/10.3389/fpsyg.2015.00966

- 52. de Wild, S. R., Simons, J. M., VranckenPeeters, M.-J.T.F.D., Smidt, M. L., Koppert, L. B., MINIMAX Group. (2022). MINImal vs MAXimal invasive axillary staging and treatment after neoadjuvant systemic therapy in node positive breast cancer: Protocol of a Dutch multicenter registry study (MINIMAX). *Clinical breast cancer*, 22(1), e59–e64. https://doi.org/10.1016/j.clbc.2021. 07.011
- Johansson, S., Svensson, H., & Denekamp, J. (2002). Dose response and latency for radiation-induced fibrosis, edema, and neuropathy in breast cancer patients. *International journal of radiation oncology, biology, physics,* 52(5), 1207–1219. https:// doi.org/10.1016/s0360-3016(01)02743-2
- Dörr, W., & Hendry, J. H. (2001). Consequential late effects in normal tissues. *Radiotherapy and oncology : Journal of the European Society for Therapeutic Radiology and Oncology*, 61(3), 223–231. https://doi.org/10.1016/s0167-8140(01)00429-7
- Straub, J. M., New, J., Hamilton, C. D., Lominska, C., Shnayder, Y., & Thomas, S. M. (2015). Radiation-induced fibrosis: Mechanisms and implications for therapy. *Journal of cancer research and clinical oncology*, *141*(11), 1985–1994. https://doi.org/10. 1007/s00432-015-1974-6
- 56. Goyal, A., Cramp, S., Marshall, A., Hammonds, N., Wheatley, D., Elsberger, B., & Dunn, J. (2022). ATNEC: a multicenter, randomized trial investigating whether axillary treatment can be avoided in patients with T1–3N1M0 breast cancer with no residual cancer in the lymph glands after neoadjuvant chemotherapy. *Journal of Clinical Oncology*, 40(16_suppl), TPS615–TPS615. https:// doi.org/10.1200/JCO.2022.40.16_suppl.TPS615
- 57 Mamounas, E. P., Bandos, H., White, J. R., Julian, T. B., Khan, A. J., Shaitelman, S. F., & Wolmark, N. (2019). NRG oncology/

NSABP B-51/RTOG 1304: phase III trial to determine if chest wall and regional nodal radiotherapy (CWRNRT) post mastectomy (Mx) or the addition of RNRT to whole breast RT post breast-conserving surgery (BCS) reduces invasive breast cancer recu. *Journal of Clinical Oncology*, *37*(15_suppl), TPS600– TPS600. https://doi.org/10.1200/JCO.2019.37.15_suppl.TPS600

- 58. Henke, G., Knauer, M., Ribi, K., Hayoz, S., Gérard, M.-A., Ruhstaller, T., & Weber, W. P. (2018). Tailored axillary surgery with or without axillary lymph node dissection followed by radiotherapy in patients with clinically node-positive breast cancer (TAXIS): study protocol for a multicenter, randomized phase-III trial. *Trials*, *19*(1), 667. https://doi.org/10.1186/s13063-018-3021-9
- 59. de Boniface, J., Frisell, J., Andersson, Y., Bergkvist, L., Ahlgren, J., Rydén, L., SENOMAC Trialists' Group. (2017). Survival and axillary recurrence following sentinel node-positive breast cancer without completion axillary lymph node dissection: the randomized controlled SENOMAC trial. *BMC Cancer*, *17*(1), 379. https://doi.org/10.1186/s12885-017-3361-y
- 60. Banys-Paluchowski, M., Gasparri, M. L., de Boniface, J., Gentilini, O., Stickeler, E., Hartmann, S., The Axsana Study Group. (2021). Surgical management of the axilla in clinically node-positive breast cancer patients converting to clinical node negativity through neoadjuvant chemotherapy: current status, knowledge gaps, and rationale for the EUBREAST-03 AXSANA Study. *Cancers*. https://doi.org/10.3390/cancers13071565

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.